Author’s response to reviews

Title: Characteristics of control group participants who increased their physical activity in a cluster-randomized lifestyle intervention trial

Authors:

Lauren A Waters (l.waters1@uq.edu.au)
Marina M Reeves (m.reeves@sph.uq.edu.au)
Brianna S Fjeldsoe (b.fjeldsoe@sph.uq.edu.au)
Elizabeth G Eakin (e.eakin@sph.uq.edu.au)

Version: 4 Date: 7 January 2011

Author’s response to reviews:

Thankyou for accepting our manuscript for publication in BMC Public Health. As requested, the trial registration number of the original RCT has now been included at the end of the abstract so that studies can be linked.

As the reviewers did not suggest further changes, no amendments have been made to the manuscript.

We look forward to sharing our research findings with the readers of BMC Public Health.

Kind regards,

Lauren Waters